Author's response to reviews

Title: Acceptability of intrapartum HIV counselling and testing in Cameroon

Authors:

Eugene J Kongnyuy (kongnyuy@gmail.com)  
Enow R Mbu (rembu2000@yahoo.com)  
F X Mbopi-Keou (fxmkeou@hotmail.com)  
Nelson N Fomulu (fomulunelson@yahoo.fr)  
Philip N Nana (njotang@yahoo.fr)  
Pierre M Tebeu (pmtebeu@yahoo.fr)  
Rebecca N Tonye (rtonye@yahoo.com)  
Robert JI Leke (robert.leke@camnet.cm)

Version: 7 Date: 11 February 2009

Author's response to reviews:

Dear Editorial Production Team,

We have addressed the comments about formatting. Please find below the comments and our responses.

Kind regards,

Eugene

Formatting changes requested

-------------------------

Major revisions (We require the author to make these changes)

Table vertical elements - Please note that we are unable to display vertical rules within the tables nor are we able to write text vertically within the cells - please re-layout your table without these vertical elements. We are unable to correctly display merged cells where the merged cell crosses rows. (Tables 1 and 2).

The tables have been formatted to remove vertical rules. Please if you find any vertical rule that might have remained in the tables, ignore the rules and treat the tables as if there were no vertical rules. Thanks.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above)

"Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any),
surname) should be written identically and in the order in which they are listed in the submission system".

The list of authors has been written in exact order they appear in the submission system. Thanks.

"Authors' contributions - please use the following format (please use initials to refer to each author’s contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript".

These changes have been made. Thanks.

"Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance".

The manuscript has been checked. Thanks.